Author's response to reviews

Title: Constrictive pericarditis in a post renal transplant subject: A case report

Authors:

Ranga M Weerakkody (rangamw2003@yahoo.com)
Harshani D Perera (harshani1983@gmail.com)
Chaminda H Kularathne (kularathnech@yhoo.com)
Mohammed HR Sheriff (rezvi.sheriff@gmail.com)

Version: 3 Date: 31 August 2013

Author's response to reviews: see over
Editorial Team,

The Journal of Medical Case Reports,

Dear Sir / Madam,

We have made the suggested changes in the manuscript. Here are the summary of changes we have made.

1. **Cover letter** removed from the manuscript
2. Renal transplant, constrictive pericarditis, tuberculosis and isoniazid prophylaxis included as **keywords**
3. **Ethnicity** of the patient added to the abstract
4. Case Report **section header** changed to Case Presentation
5. **Consent form** completed using the suggested wording.
6. **List of abbreviations** added
7. **Competing Interests** reformatted using the suggested wording.
8. **Acknowledgement**. Section updated.
9. Changes to **grammer** was made.
10. Adjustments suggested by the reviewers performed.
   a. Title changed to – Constrictive pericarditis in a post renal transplant subject:
      A case report
   b. Citation added to the text.
   c. Some adjustments in grammar made to the text.

Looking forwards for a favorable reply.

Thank you

Dr. Ranga Weerakkody.